Author's response to reviews

Title: Promoting prudent use of antibiotics: the experience from a multifaceted regional campaign in Greece

Authors:

Diamantis Plachoura (dplach@gmail.com)
Anastasia Antoniadou (ananto@med.uoa.gr)
Efthimia Giannitsioti (gianiemi@hotmail.com)
Lambrini Galani (e.giamarellou@hygeia.gr)
Ioannis Katsarolis (ikatsarolis@gmail.com)
Dimitra Kavatha (dimitra.kavatha@gmail.com)
George Koukos (georgekoukos78@yahoo.gr)
Periklis Panagopoulos (ppanag@med.uoa.gr)
Antonios Papadopoulos (antpapa1@otenet.gr)
Garyfallia Poulakou (gpoulakou@gmail.com)
Vissaria Sakka (vissaria@gmail.com)
Maria Souli (msouli@med.uoa.gr)
Styliani Symbardi (ikatsarolis@gmail.com)
Sotirios Tsiodras (sotirios.tsiodras@gmail.com)
Kyriaki Kanellakopoulou (kyrkanel@yahoo.gr)
Helen Giamarellou (e.giamarellou@hygeia.gr)

Version: 2 Date: 24 March 2013

Author's response to reviews:

Dear Editor,

Let me first on behalf of the authors thank you for the comments made in our last communication following the submission of our manuscript.

In response to the comments (also included in the revised submission) a section entitled "Ethics" has been added. We would like to add that all questionnaires and data handling the campaign were anonymous. The questionnaires filled for the RADT use survey were also anonymous and had been subjected to approval by the Ethics Committee of the University General Hospital ATTIKON. The survey for the use of RADT was performed under the approval of the regional Medical Association of Corinth. The public campaign was approved and took place under the auspices of the Prefecture of Corinth and was designed and executed in compliance with the Helsinki Declaration.

Looking forward to your reply.

With kind regards,

The authors’ team